Author’s response to reviews

Title: Nuclear hBD-1 accumulation in malignant salivary gland tumours

Authors:

Matthias Wenghoefer (m.wenghoefer@gmx.de)
Annette Pantelis (annwaldeck@gmx.de)
Henrick Dommisch (hd2@u.washington.edu)
Werner Goetz (wgoetz@uni-bonn.de)
Rudolf Reich (Rudolf.Reich@uni-bonn.de)
Markus Martini (Markus.Martini@uni-bonn.de)
Jean-Pierre Allam (Jean-Pierre.Allam@ukb.uni-bonn.de)
Soeren Jepsen (soeren.jepsen@uni-bonn.de)
Sabine Merkelbach-Bruse (Sabine.Merkelbach@uni-bonn.de)
Hans-Peter Fischer (Hans-Peter.Fischer@uni-bonn.de)
Stefaan Bergé (S.Berge@mkc.umcn.nl)
Natalija Novak (naliija.novak@ukb.uni-bonn.de)
Jochen Winter (jochen.Winter@ukb.uni-bonn.de)

Version: 7 Date: 21 September 2008

Author’s response to reviews: see over
To
Mr. Iratxe Puebla
Senior Assistant Editor
BMC-series journals

Dear Mr Pueblo,

herewith I am returning the fourth revision of the article: "MS: 2091676334166652 - Nuclear hBD-1 accumulation in malignant salivary gland tumours." my reply is in red text colour:

Major Revisions

Author list - The list of authors in the manuscript should be written exactly as they are in the submission system. The authors (forename, initial (if any), surname) should be written identically and in the order in which they are listed in the submission system.

Changed as suggested.

Corresponding author - One or more corresponding author(s) must be indicated on the title page, and their email address must be given.

Changed as suggested.
Affiliations - Each affiliation should be written in full, in the following format: department, institute, city, and country. This may mean repeating institutional addresses if the authors are from different departments within an institution. Can we also ask you to link each author to his or her corresponding affiliation with the use of superscript numbers? Please do not use full capitals for country names, and please do not include full stops in abbreviated country names.

Changed as suggested.

Highlighting/tracking- Please remove all highlighting and tracking from the manuscript.

Changed as suggested.

Table Position - Please move tables to after Reference section

Changed as suggested.

Figure legends- Figure legend X contains graphics. These will either need to be included within the image as a key, or written in text (e.g. "black square").
Figure legends - Please move these to a position after the references

Figure legends were deleted from the images and positioned after the reference section

Minor Revisions

Keywords, Word-counts, Running head and Running title - Please remove these.

Were removed.

References- Reference XX has not extracted the authors names correctly. Commas need to be inserted between the surname and forename initial. When that's done it will extract correctly, for example: Lewis B; Coates R will extract as L B; C R
If you change that to: Lewis, B; Coates, R. Then it will extract properly.
Figure cropping - It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here:

I cropped the figures to fit them into a word page.

Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions as this is the final proofing stage for your manuscript, and you will not be able to make changes after acceptance.

Yours sincerely

Dr. Dr. M. Wenghoefer